Author’s response to reviews

Title: A Comprehensive Overview on the Surgical Management of Secondary Lymphedema of the Upper and Lower Extremities related to Prior Oncologic Therapies

Authors:

Ramon Garza III (Ramongarza3@gmail.com)

Roman Skoracki (roman.skoracki@osumc.edu)

Karen Hock (karen.hock@osu.edu)

Stephen P. Povoski (stephen.povoski@osumc.edu)

Version: 1 Date: 20 May 2017

Author’s response to reviews:

May 20, 2017

Dear Dr. Dafne Solera and The BMC Cancer Editorial Office:

We are pleased to submit our revised manuscript (A Comprehensive Overview on the Surgical Management of Secondary Lymphedema of the Upper and Lower Extremities related to Prior Oncologic Therapies), as based upon the comments of Reviewer #1 and Reviewer #2.

Reviewer #1 Comments: “a well written comprehensive review paper. Perhaps authors can add a section on future of lymphatic surgery”.

Author Response to Reviewer #1: Thank you for your positive review and for you helpful suggestion. As suggested by Reviewer #1, we have added a sub-section, entitled “Future direction of lymphatic surgery”, at the end of the “Current research and the future of secondary lymphedema management” section of the Review Paper. We are hopeful that this will be appropriate to address the suggestion given by Reviewer #1.

Reviewer #2 Comments: “This is an excellent comprehensive and balanced review”.

Author Response to Reviewer #2: Thank you for your positive review.

Thank you.

Best regards,

steve
Stephen P. Povoski, M.D.
Associate Professor of Surgery
Division of Surgical Oncology
Department of Surgery
The Ohio State University Comprehensive Cancer Center - Arthur G. James Cancer Hospital and Richard J. Solove Research Institute
The Ohio State University
N-924 Doan Hall
410 West 10th Avenue
Columbus, Ohio 43210-1228
Telephone: 614-293-8700
Fax: 614-293-3465
E-mail: stephen.povoski@osumc.edu